

Fisher College of Business Working Paper Series

Charles A. Dice Center for Research in Financial Economics

Does Risk Explain Anomalies? Evidence from Expected Return Estimates

Jin (Ginger) Wu, University of Georgia

Lu Zhang, Department of Finance, The Ohio State University, NBER,

Dice Center WP 2010-18 Fisher College of Business WP 2010-03-018

October 2010

This paper can be downloaded without charge from: <u>http://www.ssrn.com/abstract=1685985</u>

An index to the working paper in the Fisher College of Business Working Paper Series is located at: http://www.ssrn.com/link/Fisher-College-of-Business.html

Does Risk Explain Anomalies? Evidence from Expected Return Estimates

Jin (Ginger) Wu^{*} University of Georgia Lu Zhang[†] Ohio State University and NBER

October 2010^{\ddagger}

Abstract

Average realized returns equal average expected returns plus average unexpected returns. If anomalies are driven by risk, average expected returns should be close to average realized returns. If anomalies are driven by mispricing, unexpected returns should be more important. We estimate accounting-based expected returns to zero-cost trading strategies formed on anomaly variables such as book-to-market, size, composite issuance, net stock issues, abnormal investment, asset growth, investment-to-assets, accruals, earnings surprises, failure probability, return on assets, and short-term prior returns. Our findings are striking. Except for the value premium, expected return estimates differ dramatically from average return estimates. The evidence suggests that mispricing, not risk, is the main driving force of capital markets anomalies.

^{*}Terry College of Business, University of Georgia, 443 Brooks Hall, Athens GA 30602. Tel: (706) 542-3638 and e-mail: jinw@terry.uga.edu.

[†]Department of Finance, Fisher College of Business, Ohio State University, 760A Fisher Hall, 2100 Neil Avenue, Columbus OH 43210; and NBER. Tel: (614) 292-8644, fax: (614) 292-7062, and e-mail: zhanglu@fisher.osu.edu.

[‡]We thank Adlai Fisher (WFA discussant), Kewei Hou, S. P. Kothari, Ravi Jagannathan, K. C. John Wei (CICF discussant), and other participants at the 2009 Western Finance Association Annual Meetings in San Diego and the 2010 China International Conference in Finance in Beijing for helpful comments. This work supersedes our working paper previously circulated under the title "Do anomalies exist ex ante?". All remaining errors are our own.

1 Introduction

We ask whether risk or mispricing is the main driving force behind capital markets anomalies, which are empirical relations between average returns and firm characteristics not explained by standard asset pricing models. Over the past three decades, anomalies have become important in asset allocation, capital budgeting, security analysis, hedge fund strategies, and many other applications. Understanding their driving forces is one of the most important questions in capital markets research.

Two competing schools of thoughts have proposed an array of economic explanations for capital markets anomalies (see Appendix A for a brief review). Behavioral finance contends that investors make systematic mistakes in pricing assets, and that anomalies are driven by predictable pricing errors (mispricing) (e.g., Barberis, Shleifer, and Vishny (1998), Daniel, Hirshleifer, and Subrahmanyam (1998), and Hong and Stein (1999)). In contrast, retaining the assumption of rational expectations, new classical finance argues that risk and expected returns vary with firm characteristics in a systematic way, and that anomalies are driven by time-varying expected returns (risk) (e.g., Cochrane (1996), Berk, Green, and Naik (1999), Zhang (2005), and Liu, Whited, and Zhang (2009)).

We aim to further the risk versus mispricing debate by estimating accounting-based expected returns to anomalies-based trading strategies. The basic idea is simple. Average realized returns equal average expected returns plus average unexpected returns. If anomalies are driven by risk, average expected returns should account for the bulk of average realized returns. If anomalies are driven by mispricing, average unexpected returns should account for the bulk of average realized returns. Building on the latest accounting literature on expected return estimation, we use the residual income model to estimate expected returns for zero-cost trading strategies formed on a comprehensive list of anomaly variables. Under the model, the expected return can be calculated as the internal rate of return that equates the present value of expected future residual incomes to the current stock price (e.g., Gebhardt, Lee, and Swaminathan (2001)). Using accounting-based models has the added advantage that accounting identities always hold regardless of investor rationality or lack thereof. Our key message is that expected return estimates differ drastically from average return estimates for most anomalies, suggesting that mispricing, not risk, is the main driving force behind anomalies. In particular, the expected return estimate of the value-minus-growth quintile is 6.3% per annum, which is close to the average return estimate of 5.2% in terms of economic magnitude. The expected return estimate, which is about 12 standard errors from zero, is also more precise than the average return estimate, which is slightly more than two standard errors from zero. The expected return estimate of the small-minus-big quintile is 3.1%, which is close to the average return estimate of 3%. The expected return estimate is more than 7.5 standard errors from zero, whereas the average return estimate is within one standard error of zero.

However, for all the other anomaly variables, the average return estimates and the expected return estimates differ dramatically in terms of economic magnitude. In the data, the high-minus-low quintiles formed on Sloan's (1996) accruals, Titman, Wei, and Xie's (2004) abnormal corporate investment, Daniel and Titman's (2006) composite issuance, and Fama and French's (2008) net stock issues all earn negative average returns, which range from -4.0% to -7.5% per annum. These average returns are at least 2.4 standard errors from zero. In contrast, the expected return estimates of these zero-cost quintiles are all between -0.1% to zero (and insignificant), meaning that these anomalies are mostly driven by unexpected returns. Although the high-minus-low quintile formed on Cooper, Gulen, and Schill's (2008) asset growth has an expected return estimate of -0.6% (t = -3.0), its magnitude does not come close to its average return estimate of -5.6% (t = -2.8).

What makes matters worse is that the expected return estimates even have the opposite signs as the average return estimates for the high-minus-low quintiles formed on Jegadeesh and Titman's (1993) prior six-month returns (momentum), Chan, Jegadeesh, and Lakonishok's (1996) earnings surprises, Campbell, Hilscher, and Szilagyi's (2008) failure probability, and Chen, Novy-Marx, and Zhang's (2010) return on assets. The high-minus-low earnings surprises, return on assets, and momentum quintiles have average returns of 4.9%, 6.5%, and 6.4% per annum, respectively, all of which are at least three standard errors from zero. However, their expected return estimates are all significantly negative, ranging from -0.1% to -1.7%. The high-minus-low failure probability quintile has an average return of -8.1% (t = -5.0), but its expected return is significantly positive, 3.8%.

We also address several recent critiques on the Gebhardt, Lee, and Swaminathan (2001) estimation of expected returns. First, because the baseline procedure uses analysts earnings forecasts that are limited to a small sample and that are likely even biased, we modify this procedure to avoid the use of analysts forecasts. Instead, we forecast future profitability using cross-sectional regressions similar to those in Fama and French (2006) (see also Hou, Dijk, and Zhang (2009)). We find that our basic inferences are robust to the use of profitability forecasts based on cross-sectional regressions.

Second, in a stream of influential articles, Easton, Taylor, Shroff, and Sougiannis (2002), Easton (2006, 2007), and Easton and Sommers (2007) criticize the Gebhardt, Lee, and Swaminathan (2001) procedure on the ground that the assumed growth rates beyond the short forecast horizon can be inconsistent with the actual growth rates in the data. This inconsistency can introduce bias in the expected return estimates. These authors propose methods that can estimate the expected returns and the expected growth rates for a portfolio simultaneously. To evaluate the impact of the growth rate assumption, we implement these alternative estimation methods on our testing portfolios. We find that these methods often predict growth rate spreads that go in the opposite direction as the growth rate spreads in the data. These counterfactual predictions cast serious doubt on the validity of these alternative methods for estimating expected returns. While we do not improve on the treatment of growth rates in the Gebhardt et al. procedure, our results suggest that their procedure is probably among the best accounting-based expected return models in the contemporary literature.

The rest of the paper is organized as follows. Section 2 describes our data and expected return estimation. Section 3 presents the expected return estimates for anomalies-based trading strategies. Section 4 deals with recent critiques on the estimation methodology. Finally, Section 5 concludes.

2 Empirical Design

We describe our data in Section 2.1 and delineate the expected return estimation in Section 2.2.

2.1 Data

The monthly data on stock returns, stock prices, and number of shares outstanding are obtained from the Center for Research in Security Prices (CRSP). We obtain stock returns with and without dividend for all NYSE, Amex, and Nasdaq stocks from CRSP. We use nonfinancial firms (excluding firms with four-digit SIC codes between 6000 and 6999) listed on the CRSP monthly stock return files and the Compustat annual industrial files from 1965 through 2008. The sample size varies across anomaly variables due to data availability. Only firms with ordinary common equity are included, meaning that we exclude ADRs, REITs, and units of beneficial interest.

Anomaly Variables

We examine an extensive list of anomaly variables. To facilitate comparison, we closely follow the prior literature in defining these variables (see Appendix B for detailed variable definitions).

Book-to-market (B/M) and size (ME). High B/M stocks earn higher average returns than low B/M stocks (e.g., Rosenberg, Reid, and Lanstein (1985), Fama and French (1993), and Lakonishok, Shleifer, and Vishny (1994)). We follow Fama and French in measuring this anomaly variable. Small firms earn higher average returns than big firms (e.g., Banz (1981)). We calculate ME as the market equity (price per share times shares outstanding) from CRSP.

Composite issuance (CI) and net stock issues (NSI). Firms that issue new equity underperform, and firms that buy back shares outperform matching firms with similar characteristics in the future three to five years (e.g., Ritter (1991), Loughran and Ritter (1995), Ikenberry, Lakonishok, and Vermaelen (1995), and Michaely, Thaler, and Womack (1995)). We use two variables to summarize the external financing anomalies. From Daniel and Titman (2006), CI measures the part of firm growth in market equity that is not due to stock returns. From Fama and French (2008), *NSI* measures the annual change in the logarithm of the number of real shares outstanding, which adjusts for distribution events such as splits and rights offerings.

Abnormal investment (AI), asset growth (AG), investment-to-assets (I/A), and accruals (AC). Titman, Wei, and Xie (2004) show that firms with abnormally high investment earn lower average returns than firms with abnormally low investment. AI is the deviation of the current year's investment from the benchmark investment, which is defined as the past three-year moving average of investment. Cooper, Gulen, and Schill (2008) show that firms with high asset growth earn lower average returns than firms with low asset growth. AG is measured as the annual percentage change in total assets. Lyandres, Sun, and Zhang (2008) and Chen, Novy-Marx, and Zhang (2010) show that high I/A firms earn lower average returns than low I/A firms. I/A is the annual change in gross property, plant, and equipment (Compustat annual item PPEGT) plus the annual change in inventory (item INVT) divided by the lagged total assets (item AT). Sloan (1996) shows that high AC firms earn lower average returns than low AC firms. Following Sloan, we measure AC as changes in non-cash working capital minus depreciation expense scaled by average total assets.

Standardized Unexpected Earnings (SUE) and return on assets (ROA). High SUE stocks earn higher average returns than low SUE stocks (e.g., Ball and Brown (1968), Bernard and Thomas (1989), and Chan, Jegadeesh, and Lakonishok (1996)). The definition of SUE for stock *i* in month *t* is $(e_{iq} - e_{iq-4})/\sigma_{it}$, where e_{iq} is the most recently announced quarterly earnings per share (Compustat quarterly item EPSPIQ) as of month *t* for stock *i*, e_{iq-4} is earnings per share announced four quarters ago, and σ_{it} is the volatility of $e_{iq} - e_{iq-4}$ over the prior eight quarters. Chen, Novy-Marx, and Zhang (2010) show that high ROA firms earn higher average returns than low ROA firms. We measure return-on-assets, ROA, as income before extraordinary items (Compustat quarterly item IBQ) divided by last quarterly's assets (item ATQ).

Failure probability (FP). The financial distress anomaly says that more distressed firms earn abnormally lower average returns than less distressed firms (e.g., Dichev (1998) and Campbell,

Hilscher, and Szilagyi (2008)). Following Campbell et al., we measure distress as a linear function of the ratio of earnings over the market value of the firm, monthly excess return relative to the S&P 500 index, market leverage, stock return volatility, relative size, the ratio of cash over the market value of the firm, market-to-book equity, and log price per share.

Momentum (MOM). Jegadeesh and Titman (1993) show that stocks that perform well in the recent six to twelve months continue to earn higher average returns in the future six to twelve months than stocks that perform poorly in the recent six to twelve months. Following Jegadeesh and Titman, we measure momentum as prior six-month returns.

Portfolio Construction

We construct one-way quintile portfolios based on the anomaly variables. In June of each year t from 1965 to 2008, we sort all NYSE stocks on CRSP on book-to-market, size, composite issuance, net stock issues, abnormal investment, asset growth, investment-to-assets, and accruals. We use the NYSE breakpoints to split NYSE, Amex, and Nasdaq stocks into one-way quintiles, and calculate annual value-weighted returns from July of year t to June of year t + 1. Firms with negative book equity for the fiscal year ending in calendar year t - 1 are excluded.

For each month from January 1977 to December 2008, we sort all NYSE stocks on their most recent *SUEs*, and use the NYSE breakpoints to split NYSE, Amex, and Nasdaq stocks into five groups. We hold the resulting portfolios for six months, and calculate value-weighted returns. The sample starts from January 1977 due to the availability of quarterly earnings data.

Following Campbell, Hilscher, and Szilagyi (2008), for each month from January 1975 to December 2008, we sort all NYSE, Amex, and Nasdaq stocks on CRSP on failure probability into five groups. We use Compustat accounting data for a fiscal quarter in portfolio sorts in the months immediately after the quarter's public earnings announcement dates (Compustat quarterly item RDQ). We calculate the one-year buy-and-hold value-weighted returns of stocks with and without dividends for each portfolio. The starting period of the sample is restricted by the availability of quarterly data on total liabilities in the definition of failure probability.

To construct the ROA quintiles, we sort NYSE stocks based on the ranked values of quarterly ROA, and use the NYSE breakpoints to split NYSE, Amex, and Nasdaq stocks into quintiles. We use quarterly earnings in portfolio sorts only in the months immediately after the most recent earnings announcement (Compustat quarterly item RDQ). For example, if the earnings for the fourth fiscal quarter in year t are announced on March 5 (or March 25) of year t + 1, we use the announced earnings to calculate ROA to form portfolios at the beginning of April and to calculate the resulting portfolio returns over April of year t+1. In particular, monthly value-weighted returns on the quintiles are calculated for the current month, and the portfolios are rebalanced monthly.

Finally, Following Jegadeesh and Titman (1993), for each month from July 1965 to June 2008, we sort all NYSE stocks on CRSP on the prior six-month returns and use the NYSE breakpoints to split NYSE, Amex, and Nasdaq stocks into quintiles. We hold the portfolios for six months, and calculate the value-weighted returns with and without dividends.

2.2 Expected Return Estimation

The estimation method is from Gebhardt, Lee, and Swaminathan (2001, GLS hereafter). GLS compute the expected return as the internal rate of return (implied cost of equity) that equates the present value of expected future cash flows in the residual income model to the current stock price.¹

The Baseline Procedure

We closely follow GLS's procedure in our baseline estimation. We use the analyst earnings forecasts from Institutional Brokers' Estimate System (IBES) as the proxy for the market's earnings

¹A large literature in accounting uses valuation models to estimate expected returns. Many studies calculate expected returns from analysts earnings forecasts under the residual income model (e.g., Claus and Thomas (2001), Gebhardt, Lee, and Swaminathan (2001), Gode and Mohanram (2003), Guay, Kothari, and Shu (2005), Hou, Dijk, and Zhang (2009), and Lee, So, and Wang (2010)). Easton, Taylor, Shroff, and Sougiannis (2002), Easton (2006), and Easton and Sommers (2007) implement the residual income model by estimating the expected returns and the implied growth rates simultaneously at the portfolio level (see Section 4 for more discussion on these methods). Francis, LaFond, Olsson, and Schipper (2004) and Brav, Lehavy, and Michaely (2005) use Value Line analysts expectations to estimate expected returns. Pastor, Sinha, and Swaminathan (2008), Lee, Ng, and Swaminathan (2009), and Chava and Purnanandam (2010) apply these estimation methods to address important questions in finance.

expectations. We compute a finite horizon estimate of equity value for each firm:

$$P_t = B_t + \frac{FROE_{t+1} - E_0[R]}{1 + E_0[R]}B_t + \frac{FROE_{t+2} - E_0[R]}{(1 + E_0[R])^2}B_{t+1} + TV,$$
(1)

in which $E_0[R]$ denotes the expected return estimate from the baseline estimation. B_t is the book value from the most recent financial statement divided by the number of shares outstanding in the current month. $FROE_{t+\tau}$ is forecasted return on equity (ROE) for period $t + \tau$. For the first three years, we compute it as $FEPS_{t+\tau}/B_{t+\tau-1}$, in which $FEPS_{t+\tau}$ is the mean forecasted earnings per share (EPS) for year $t + \tau$ from IBES, and $B_{t+\tau-1}$ is the book value per share for year $t + \tau - 1$.

We use the mean analysts' one-year and two-year ahead earnings forecasts ($FEPS_{t+1}$ and $FEPS_{t+2}$, respectively) and the long-term growth rate estimate (Ltg) from IBES to compute the three-year-ahead earnings forecast as $FEPS_{t+3} = FEPS_{t+2}(1 + Ltg)$. Beyond the third year, we forecast FROE using a linear interpolation to the industry median ROE. To calculate the industry median ROE, we sort all stocks into the 48 industries classified by Fama and French (1997). The industry median ROE is the ten-year (at least five-year) moving median of past ROEs of all firms in the industry. Loss firms are excluded from the calculation of the industry median.

Book equity per share is $B_{t+\tau} = B_{t+\tau-1} + FEPS_{t+\tau} - FDPS_{t+\tau}$, in which $FDPS_{t+\tau}$ is the forecasted dividend per share for year $t+\tau$, estimated using the current dividend payment ratio (k =dividends for the most recent fiscal year divided by earnings over the same time period, $0 \le k \le 1$), i.e., $FDPS_{t+\tau} = k \times FEPS_{t+\tau}$. For firms with negative earnings we divide the dividends by 0.06 times total assets to derive an estimated payout ratio. Payout ratios of less than zero are assigned a value of zero, and payout ratios greater than one are assigned a value of one. We forecast earnings up to 12 future years and estimate a terminal value TV for cash flows beyond year 12:

$$TV = \sum_{i=3}^{T-1} \frac{FROE_{t+\tau} - E_0[R]}{(1+E_0[R])^i} B_{t+\tau-1} + \frac{FROE_{t+T} - E_0[R]}{E_0[R](1+E_0[R])^{T-1}} B_{t+T-1}.$$
 (2)

We estimate the implied cost of equity, $E_0[R]$, for each firm in each month by substituting the

forecasted future earnings, book values, and terminal values into equation (1) and solving for $E_0[R]$ from the resulting nonlinear equation. For portfolios that are annually rebalanced at the end of June of year t, we value-weight $E_0[R]$ measured at the end of December of year t-1 across firms in each testing portfolio to obtain portfolio-level expected returns. This timing convention means that we match the expected returns at the end of year t-1 with expost returns from July of year t to June of year t+1. The six-month lag between January and June of year t is imposed per Fama and French (1993) to allow accounting information to be released to the market.

For the monthly rebalanced momentum portfolios, for each month we sort all NYSE stocks on CRSP on the prior six-month realized returns and use the NYSE breakpoints to split NYSE, Amex, and Nasdaq stocks into quintiles. We hold the portfolios for six months and value-weight the expected returns across firms in a given portfolio for each month. Although $E_0[R]$ is available monthly because P_t and $FEPS_t$ are updated monthly, $E_0[R]$ is the expected future one-year return. The procedure for the SUE portfolios is similar. For each month we sort all NYSE stocks on their most recent past SUE, and use the NYSE breakpoints to split NYSE, Amex, and Nasdaq stocks into quintiles. We hold the resulting portfolios for six months and calculate the value-weighted $E_0[R]$ estimated for each month. For the monthly rebalanced ROA portfolios, we use NYSE breakpoints to sort all stocks into quintiles based on the most recent ROA at the beginning of each month. For the FP quintiles, we sort all NYSE, Amex, and Nasdaq stocks on the most recent FP into quintiles in each month. We calculate the value-weighted $E_0[R]$ for each portfolio in each month.

Two Modified Estimation Procedures

The baseline estimation of the implied costs of equity uses analysts earnings forecasts from IBES as expected earnings. Two potential issues arise with this procedure in our application. First, analysts earnings forecasts tend to be overly optimistic (e.g., O'Brien (1988)), and as a result, expected return estimates implied by these forecasts tend to be upward biased (e.g., Easton and Sommers (2007)). If this bias varies systematically with anomaly variables (for example, analysts might be more optimistic toward growth firms, high accrual firms, and firms that issue equity), the estimates of expected returns to zero-cost strategies will also be biased. Second, because analysts tend to follow larger, more visible stocks, expected return estimates are limited to a small sample of stocks that have analysts coverage. This limitation can affect the results for anomalies-based trading strategies that often involve stocks that are not followed by analysts.

To address these issues, we use two modified procedures for estimating implied costs of equity. The baseline approach uses analysts earnings forecasts in forming forecasted return on equity, $FROE_{t+\tau}$. We instead forecast future one-, two-, and three-year ahead ROEs using cross-sectional regressions similar to those in Fama and French (2006). Specifically, we estimate Fama-MacBeth (1973) cross-sectional regressions of future realized $ROE_{t+\tau} = Y_{t+\tau}/B_{t+\tau-1}$, in which $\tau = 1, 2, 3$, and $Y_{t+\tau}$ is τ -year ahead realized earnings per share. (Fama and French forecast $Y_{t+\tau}/B_t$, but we forecast $Y_{t+\tau}/B_{t+\tau-1}$ to provide inputs into the implied costs of equity estimation.)

In the first modified procedure, we use Fama and French's (2006) full specification, including the logarithm of book-to-market, the logarithm of market equity, a dummy variable that is one for firms with negative earnings for fiscal year t (zero otherwise), Y_t/B_t , $-AC_t/B_t$ with $-AC_t$ being accruals per share for firms with negative accruals (zero otherwise), $+AC_t/B_t$ with $+AC_t$ being accruals per share for firms with positive accruals (zero otherwise), asset growth for fiscal year t, a dummy variable that is one for firms that pay no dividends for fiscal year t, and the ratio of dividends to book equity. The full list of predictors imposes data requirements such that the resulting sample size is similar to that in the baseline procedure. To enlarge the sample size, in the second modified procedure we use a simplified list of predictors to forecast ROE, including only the log book-to-market, the log market equity, the negative earnings dummy, Y_t/B_t , and the current asset growth. To avoid look-ahead bias, we use ten-year rolling windows (at least five years) up to year t to forecast future ROE.

Because we forecast ROE directly, as opposed to earnings per share, the baseline estimation of the implied costs of equity needs to be adjusted accordingly. To compute future book equity per share, we still use the clean surplus relation: $B_{t+\tau} = B_{t+\tau-1} + (1-k) \times FEPS_{t+\tau}$, in which k is the dividend payout ratio. However, the forecasted earnings per share $FEPS_{t+\tau}$ is calculated as $FROE_{t+\tau} \times B_{t+\tau-1}$, in which $FROE_{t+\tau}$ with $\tau = 1, 2, 3$ is the forecasted ROE from the cross-sectional regressions. All other aspects of the estimation procedure remain the same as in the baseline procedure. Our modified procedures are in the same spirit as Hou, Dijk, and Zhang (2009), who use cross-sectional regressions to forecast the earnings of individual firms. However, because earnings might appear nonstationary, we opt to forecast ROE directly. Comparing the estimates across the baseline and modified procedures can shed light on whether biases in analysts earnings forecasts cause any bias in the expected returns to anomalies-based trading strategies.

Descriptive Statistics

Panel A of Table 1 reports the descriptive statistics for the sample used in the baseline implied costs of equity estimation. Because doing so requires analysts earnings forecasts from IBES, the average numbers of firms in the cross-section for the B/M, CI, and AI quintiles are only 2,201, 1,393, and 1,513, respectively. Panel B reports the descriptive statistics for the sample used in the implied costs of equity estimation in which we use the full ROE forecasting regressions from Fama and French (2006). Although this procedure is immune to analysts forecasting bias, the sample size is comparable with that based on IBES. In particular, the average numbers of firms in the crosssection for the B/M, CI, and AI quintiles are 2,091, 1,134, and 1,540, respectively. The reason is that the full Fama-French specification requires firms to have nonmissing observations for many forecasting variables simultaneously. To increase the sample size, we also implement the simplified Fama-French ROE forecasting regressions with a shorter list of variables. Panel C shows that doing so substantially increases the sample size relative to that in Panel B. The average numbers of firms in the cross-section for the B/M, CI, and AI quintiles increase to 2,893, 1,534, and 2,025, respectively.

3 Expected Return Estimates as Implied Costs of Equity

After we discuss intermediate results on forecasting profitability in Section 3.1, we turn to our central results on expected return estimation in Section 3.2.

3.1 Forecasting Profitability

Table 2 reports the average slopes and their *t*-statistics for annual cross-sectional profitability forecasting regressions using the Fama-MacBeth (1973) methodology. We report the regression results from the full sample (although as noted, we use ten-year rolling windows to estimate the crosssectional regressions when estimating implied costs of equity to guard against look-ahead bias).

Lagged ROE is the strongest predictor of future ROE. In the full specification, the average slope on lagged ROE for one-year ahead ROE is 0.63, which is more than 18 standard errors from zero. The evidence shows considerable persistence in the ROE. The slope decays to 0.39 in forecasting three-year ahead ROE, which is still more than 13 standard errors from zero. The evidence from the short specification is similar. The average slope on lagged ROE for one-year ahead ROE is 0.61, which is more than 18 standard errors from zero. Size forecasts future ROE with significantly positive slopes, meaning that big firms are more profitable than small firms. For the most part, B/M forecasts ROE with significantly negative slopes. As such, growth firms are more profitable than value firms. Firms that do not pay dividends are less profitable than firms that do pay dividends. Firms with high dividends to book equity ratios are more profitable than firms with low dividends to book equity ratios. The evidence is largely consistent with Fama and French (2006).

3.2 Expected Return Estimates

Table 3 reports the key message of the paper. The table shows the expected returns for all the testing portfolios from the implied costs of equity estimation. To facilitate comparison, we also report the average realized returns for the testing portfolios in the sample used for the baseline estimation. Despite the fact that the IBES sample tilts toward big firms, the magnitude of the

anomalies measured with average realized returns in the IBES sample are largely similar to those in a broad sample without restricting firms to be covered by IBES (not reported). To preview the result, Table 3 shows that for most anomalies, the average return and the expected return estimates differ dramatically across the testing portfolios. This evidence suggests that most anomalies are driven by predictable unexpected returns (mispricing), as opposed to time-varying expected returns (risk).

Panel A shows that the expected value premiums from different estimation methods are similar in magnitude, and are all significantly positive. In the baseline procedure, the value quintile earns a higher expected return than the growth quintile: 14.9% versus 8.6% per annum. The spread of 6.3% is 12 standard errors from zero. This expected return spread is close to the average return spread of 5.2% across the book-to-market quintiles. However, the precision of the expected return estimate is substantially higher than that of the average return spread, which is only 2.2 standard errors from zero. In the two modified implied costs of equity procedures, the estimates of the expected value premium are both 8.5%, which are also (relatively) close to the average return estimate. Both of the expected return estimates are more than eight standard errors from zero. From Panel B, the expected return estimates of the small-minus-big quintile range from 1.8% to 3.1% per annum, which are close to the average return estimate of 3%. However, while the average return estimate is insignificant, the expected return estimates are all more than five standard errors from zero.

The similarity between average return and expected return estimates ceases to exist for the rest of the anomaly variables. From Panel C, the high-minus-low CI quintile earns an average return of -4.2%, which is more than 2.5 standard errors from zero. In contrast, the expected return estimates are substantially lower in magnitude, ranging from -0.1% to -1.1%. Although the estimates from the modified procedures are significant, the estimate from the baseline procedure is not. The results for the NSI and AI portfolios are largely similar to those for the CI portfolios. The three anomaly variables produce significantly negative average returns for the high-minus-low portfolios, but their expected return estimates are economically small and often statistically insignificant. In particular, Panel D shows that the high-minus-low NSI quintile earns an average return of -7.5% per annum, which is three standard errors from zero. In contrast, the expected return estimates of this zero-cost quintile range from -0.1% to -0.5%, which are all within 1.9 standard errors from zero. From Panel E, the high-minus-low AI quintile earns an average return of -4.0%, which is more than 2.4 standard errors from zero. However, the expected return estimates of this zero-cost quintile range from -0.1% to -1.0%, and two of three estimates are insignificant.

For the AG and I/A portfolios, although the expected return estimates of the high-minus-low quintiles are significantly negative, their magnitude is substantially lower than that of the average return estimates. From Panel F, the average return of the high-minus-low AG quintile is -5.6% per annum, which is 2.8 standard errors from zero. However, the expected return estimates range from -0.6% to -1.5%, albeit significant. Panel G shows that the average return of the high-minus-low I/A quintile is -3.4% (t = -1.8). In contrast, the expected return estimates only fall in the range between -0.5% and -1.0%, and do not come close to matching the magnitude of the average return.

From Panel H, the high-minus-low AC quintile earns an average return of -4.8% (t = -3.6). The baseline procedure yields a slightly negative expected return estimate. The two modified procedures yield expected return estimates of -0.5% and -0.3%. Although at least marginally significant, these estimates are substantially lower in magnitude than the average return estimate. Wu, Zhang, and Zhang (2010) also document that the expected return spread across the extreme accrual quintiles is too small in magnitude relative to the average return spread. Their estimates are based only on the baseline implied costs of equity estimation. We show that the expected return estimates from the modified procedures are largely similar to those from the baseline procedure, meaning that bias in analysts forecasts is not important for estimating expected returns of the accrual portfolios.

The remaining four panels in Table 3 report that the expected return estimates for the high-minus-low quintiles formed on earnings surprises, failure probability, return on assets, and momentum deviate even more from their average return estimates. In particular, the expected returns and the average returns have the opposite signs. The high-minus-low earnings surprises

quintile earns an average return of 4.9% per annum, which is 5.5 standard errors from zero. In contrast, the expected return estimates range from -0.1% to -1.0%, and are all significant. The average return of the high-minus-low failure probability quintile is -8.1%, which is five standard errors from zero. However, the baseline estimation shows that its expected return is positive, 3.8%, and is highly significant. This evidence is consistent with Chava and Purnanandam (2010), who also show that more distressed firms have higher implied costs of equity than less distressed firms in the baseline GLS estimation. We add to their work by showing that their inferences are robust to their use of analysts earnings forecasts because the two modified procedures deliver largely similar results.

From Panel K, the high-minus-low ROA quintile earns an average return of 6.5% per annum, which is 3.3 standard errors from zero. In contrast, the expected return estimate from the baseline procedure is -1.7%, which is highly significant. The estimates from the two modified procedures are largely similar: -2.1% and -2.7%, which are again highly significant. Finally, Panel L shows that the winner-minus-loser quintile earns an average return of 6.4% (t = 3.3). In contrast, the expected return estimates for the winner-minus-loser quintile range from -1.7% to -2.3%, which are all at least 17 standard errors from zero.

In summary, the central message from Table 3 is clear. The average return estimates and the expected return estimates are drastically different across the testing portfolios, except for the value premium. This evidence means that mispricing, not risk, is the main driving force behind most anomalies. We also show that the expected return estimates from the modified estimation procedures are largely similar to those from the baseline procedure. This evidence means that bias in analysts forecasts is not quantitatively important for estimating expected returns at the portfolio level.

4 Estimating Expected Returns and Expected Growth Rates Simultaneously

The basic inferences are based on the GLS procedure of estimating expected returns. Easton, Taylor, Shroff, and Sougiannis (2002, ETSS hereafter), Easton (2006, 2007), and Easton and Sommers (2007) argue that the assumed growth rates beyond the short forecast horizon in the GLS procedure can be different from the growth rates in the data, and that this difference can introduce bias in the expected return estimates. These authors propose methods to estimate the expected returns and expected growth rates of a given portfolio simultaneously. To evaluate the impact of the growth rate assumption on our basic inferences, we implement these alternative methods on our testing portfolios. We find that these methods often yield counterfactual implications for the cross-section of expected growth. In our view these results invalidate the use of these alternative methods. While these results do not improve on the treatment of growth rate in the GLS estimation, they do suggest that the GLS procedure is probably among the best that the latest accounting literature has to offer.

4.1 Methodology

To describe these alternative methods, we start with the residual income model:

$$V_{it} = B_{it} + \sum_{\tau=1}^{\infty} \frac{Y_{it+\tau} - r_i \times B_{it+\tau-1}}{(1+r_i)^{\tau}}$$
(3)

in which V_{it} is the intrinsic value per share of firm *i* at time *t*, B_{it} is book value per share, Y_{it} is earnings per share, and r_i is the cost of equity.

The Baseline ETSS Estimation

ETSS operationalize the residual income model by assuming that (starting from the period from t to t+1) the residual earnings as a perpetuity grows at a constant annual rate of g_i . This assumption means that we can reformulate equation (3) as:

$$P_{it} = B_{it} + \frac{Y_{it+1}^{IBES} - r_i \times B_{it}}{r_i - g_i}$$
(4)

in which P_{it} is price per share of firm *i* at time *t*, Y_{it+1}^{IBES} is the IBES analysts forecasts (known at time *t*) of earnings for time t + 1, and g_i is the expected growth rate in residual income beyond time t + 1 required to equate $P_{it} - B_{it}$ and the present value of the infinite residual income stream.

Some algebra shows that equation (4) is equivalent to:

$$\frac{Y_{it+1}^{IBES}}{B_{it}} = g_i + \frac{P_{it}}{B_{it}}(r_i - g_i)$$
(5)

We follow ETSS and implement this equation using Fama-MacBeth (1973) cross-sectional regressions across all the firms within a given portfolio:

$$\frac{Y_{it+1}^{IBES}}{B_{it}} = \gamma_0 + \gamma_1 \frac{P_{it}}{B_{it}} + \mu_{it} \tag{6}$$

where $\gamma_0 = g$ with g being the implied (average) growth rate for the portfolio, and $\gamma_1 = r - g$ with r being the expected return for the portfolio. We call this procedure the baseline ETSS estimation.

The Modified ETSS Estimation

Following the same idea as in the modified procedures for estimating implied costs of equity, we also replace the left-hand side of equation (6) with the forecasted one-year ahead ROE from the Fama-French (2006) ROE forecasting regressions. Doing so includes the sample observations not covered by analysts and avoids potential bias in analysts forecasts. We call this procedure the modified ETSS estimation. We use the forecasted ROE from the full Fama-French profitability regressions. Using the simplified specification yields largely similar results (not reported).

The O'Hanlon-Steele Estimation

O'Hanlon and Steele (2000) and Easton (2006) reformulate equation (3) in a different way:

$$P_{it} = B_{it} + \frac{(Y_{it} - r_i \times B_{it-1})(1 + g'_i)}{r_i - g'_i}$$
(7)

in which g'_i is the perpetual growth rate starting from the current period's residual income for the period from t-1 to t. (In contrast, g_i in equation (4) is the implied perpetual growth rate starting from the next period's residual income from t to t+1.) The implied growth rate, g'_i , produces a residual income stream such that the present value of this stream equals the difference between P_{it} and B_{it} .

Some algebra shows that equation (7) is equivalent to:

$$\frac{Y_{it}}{B_{it-1}} = r_i + \frac{r_i - g'_i}{1 + g'_i} \frac{P_{it} - B_{it}}{B_{it-1}}$$
(8)

We follow O'Hanlon and Steele (2000) and Easton (2006) and implement this equation with the following cross-sectional regression for a portfolio of stocks:

$$\frac{Y_{it}}{B_{it-1}} = \delta_0 + \delta_1 \frac{P_{it} - B_{it}}{B_{it-1}} + \mu_{it}$$
(9)

where $\delta_0 = r$ with r being the portfolio-level expected return and $\delta_1 = (r - g')/(1 + g')$ with g' being the expected growth rate for the portfolio. We call this estimation the O'Hanlon-Steele procedure.

We estimate annual value-weighted Fama-MacBeth (1973) cross-sectional regressions in each period using the Weighted Least Squares with the weights given by market capitalization. We use value-weights to facilitate comparison with the results from the implied costs of equity estimation. We implement the estimation procedures for all testing quintile portfolios. To test whether a given high-minus-low quintile has an average return of zero, we estimate the cross-sectional regressions for the two extreme quintiles in question jointly, and test the null hypothesis using the Fama-MacBeth standard errors for the implied expected returns of the high-minus-low quintile. The test on whether a given high-minus-low quintile has an implied growth rate of zero is defined analogously.

4.2 Estimation Results

Preliminaries

Panel A of Table 4 reports the descriptive statistics for the sample for the baseline ETSS estimation. The average numbers of firms in the cross-section for the B/M, CI, and AI quintiles reduce to 3,026, 1,649, and 1,753, respectively. Panel B reports the results for the sample used in the modified ETSS estimation in which we use the full ROE forecasting regressions from Fama and French (2006). Although this estimation is not subject to analysts forecasting bias, the sample size is comparable with that based on IBES in the baseline ETSS procedure. The average numbers of firms in the cross-section for the B/M, CI, and AI quintiles are 2,851, 1,507, and 1,859, respectively. Panel C describes the sample for the O'Hanlon-Steele estimation. Because this procedure does not use IBES or require a long list of variables to be available to forecast ROE, the sample size is larger. In particular, the average numbers of firms in the cross-section for the B/M, CI, and AI quintiles increase to 3,369, 1,749, and 1,983, respectively.

Expected Growth Rate Estimates

Table 5 reports the estimated growth rates from the baseline ETSS method that uses the IBES earnings forecasts, g_0 , the growth rates from the modified ETSS method that uses the Fama-French (2006) *ROE* forecasts, g_1 , and the growth rates from the O'Hanlon-Steele method, g_2 . To facilitate comparison, we also report the average dividend growth rates in the data, A[G]. We follow Hansen, Heaton, and Li (2005) and Chen, Petkova, and Zhang (2008) in measuring dividend growth rates at the portfolio level (see Appendix C for details). We find that implied growth rate spreads across the testing portfolios often go in the opposite direction as those in the data. This counterfactual pattern casts doubt on the validity of the alternative methods of estimating expected returns.

From Panel A, value firms have higher growth rates on average than growth firms in the data: 9.1% versus 5.0% per annum. The spread of 4.1% is 1.3 standard errors from zero. However, the alternative methods all predict that value firms have significantly lower expected growth rates than growth firms. In particular, the baseline ETSS procedure generates a negative growth rate spread of -4.9% for the high-minus-low B/M quintile, which is more than 2.5 standard errors from zero. The modified ETSS and the O'Hanlon-Steele procedures produce even larger spreads, -15.4% and -13.9%, respectively, which are at least ten standard errors from zero. From Panel B, small firms grow faster than big firms in the data, although the growth rate spread of 5.2% per annum is insignificant. However, the alternative methods all predict that big firms grow faster than small firms. The baseline and modified ETSS procedures predict that big firms grow faster than small firms by 5%, which is at least four standard errors from zero. The O'Hanlon-Steele procedure implies that big firms grow faster than small firms by 10.9%, which is about 4.5 standard errors from zero.

The expected growth spreads are small, positive, and insignificant for the high-minus-low quintiles formed on AI, AG, and I/A in the data. However, the alternative methods often produce significantly positive implied growth rate spreads. In particular, the high-minus-low AI quintile in the data has a growth rate of 1.2% per annum, which is within 0.5 standard errors of zero. However, the modified ETSS method implies a growth rate of 8.6%, which is more than six standard errors from zero. The implied growth rate spread from the O'Hanlon-Steele method is even higher, 15.5%, which is more than nine standard errors from zero. The growth rate spreads are small, negative and insignificant for the high-minus-low quintiles formed on CI and AC. However, the ETSS methods often produce significantly positive growth rate spreads. The high-minus-low CIquintile has a growth rate of -1.1% per annum in the data, which is within 0.5 standard errors of zero. However, the baseline ETSS method implies a growth rate spread of 9.2% (t = 3.7). The high-minus-low NSI quintile has an economically large growth rate of -6.6%, albeit insignificant. However, the baseline ETSS estimation yields an expected growth rate of 3.6% (t = 2.1).

The expected growth rate spreads across the extreme SUE quintiles are mixed. The baseline ETSS method implies a growth rate spread of 2.2% per annum, which is more than 4.5 standard errors from zero. However, the two related methods imply growth rate spreads of -1.8% and -1.3%that are more than four standard errors from zero. For comparison, the high-minus-low SUE quintile has a positive growth rate of 4.1% in the data (t = 7.6). The high-minus-low momentum quintile has a growth rate of 6.5% (t = 7.3) in the data. The implied growth rates for this portfolio from the ETSS methods are all positive, but the magnitudes range from 0.4% to 3.3%, which are lower than the observed growth rate. The high-minus-low FP quintile has a large positive growth rate of 17.8% per annum, which is more than four standard errors from zero. In contrast, the ETSS methods all forecast strongly negative implied growth rates around -20%, which are all more than 15 standard errors from zero. The observed and the implied growth rate of 5.4% in the data (t = 4.6), while the ETSS methods forecast significantly positive growth rates ranging from 9.5% to 13.2%.

Why do these alternative methods deliver counterfactual implications of expected growth? In our view the inherent specification errors within these methods are the likely culprits. For example, the cross-sectional regression in equation (6) is derived under strong assumptions. The regression assumes that there are measurement errors in Y_{it+1}^{IBES} and P_{it}/B_{it} and specification errors in equation (5). Specification errors can arise from two sources. First, the residual earnings might not be a perpetuity that grows at a constant rate. Second, P_{it}/B_{it} and $r_i - g_i$ might be correlated cross-sectionally, meaning that the average of $r_i - g_i$ cannot be treated as a constant slope in the cross-sectional regression. The ETSS procedure assumes that all these errors have a mean of zero, meaning that equation (5) can be estimated using linear cross-sectional regressions.

The cross-sectional regression in equation (9) also involves strong assumptions. In particular, specification errors can arise from three sources. First, the residual earnings might not be a perpetuity that grows at a constant rate. Second, $(P_{it} - B_{it})/B_{it-1}$ and $(r_i - g'_i)/(1 + g'_i)$ might be correlated cross-sectionally, so that the average of $(r_i - g'_i)/(1 + g'_i)$ cannot be treated as a constant slope in the cross-sectional regression. Third, because $(r_i - g'_i)/(1 + g'_i)$ is nonlinear in r_i and g'_i , Jensen's inequality means that the average of $(r_i - g'_i)/(1 + g'_i)$ cannot be replaced with $(r - g')/(1 + g'_i)$. The O'Hanlon-Steele procedure assumes that all these errors have a mean of zero, so that equation (8) can be transformed into the cross-sectional regression in equation (9).

Expected Return Estimates

Table 6 reports expected return estimates using the alternative methods that determine expected returns and growth rates simultaneously for the testing portfolios. Because of the counterfactual results on expected growth rates reported in Table 5, it is not surprising that the expected return estimates diverge from those obtained from the GLS procedure as well as average returns. Easton, Taylor, Shroff, and Sougiannis (2002) show that their baseline procedure works well at the aggregate level in that it delivers reasonable equity premium estimates. Table 6 shows that the ETSS proce-

dure fails to provide reasonable expected return estimates for anomalies-based trading strategies.

Panel A shows that the average return of the high-minus-low B/M quintile is 4.2% per annum (t = 1.7) in the sample for the baseline ETSS procedure. Unlike the positive average return, the expected return estimates are all negative: -2.1% (t = -1.0) from the baseline ETSS estimation, -12.7% (t = -12.2) from the modified ETSS estimation, and -9.4% (t = -6.0) from the O'Hanlon-Steele estimation. Panel B shows that the average return of the small-minus-big quintile is 2.2% per annum, which is within one standard error of zero. In contrast, the expected return estimates from the baseline and modified ETSS procedures are -6.4% and -7.5%, respectively, which are both more than 5.5 standard errors from zero. The estimate from the O'Hanlon-Steele procedure is -11.8%, which is more than 4.5 standard errors from zero. From Panel C, although the average return of the high-minus-low CI quintile is significantly negative, the expected return estimate from the baseline ETSS procedure is significantly positive, 5.2%, which is three standard errors from zero.

Similar drastic differences between average returns and expected returns are also evident for the NSI, AI, AG, I/A, and AC quintiles. The high-minus-low AI and I/A quintiles both earn insignificantly negative average returns. However, the modified ETSS procedure and the O'Hanlon-Steele procedure both produce significantly positive expected return estimates, which are more than six standard errors from zero. The high-minus-low AG and AC portfolios both earn significantly negative average returns. However, the modified ETSS procedure and the O'Hanlon-Steele procedure both show significantly positive expected return estimates, which are more than five standard errors from zero. The baseline ETSS procedure generates insignificant expected return estimates for the high-minus-low quintiles formed on all these anomaly variables. Without going through the details, we observe from the remaining four panels of Table 6 that the average return estimates also diverge from the expected return estimates for the rest of the anomaly variables.

5 Conclusion

We use valuation models to estimate expected returns to anomalies-based trading strategies formed on book-to-market, size, composite issuance, net stock issues, abnormal investment, asset growth, investment-to-assets, accruals, standardized unexpected earnings, failure probability, return on assets, and short-term prior returns. The central message is that except for the value premium, expected return estimates differ dramatically from average realized returns. The evidence suggests that mispricing, not risk, is the main driving force behind most capital markets anomalies.

References

- Ball, Ray, and Philip Brown, 1968, An empirical evaluation of accounting income numbers, *Journal* of Accounting Research 6, 159–178.
- Banz, Rolf W., 1981, The relationship between return and market value of common stocks, *Journal* of Financial Economics 9, 3–18.
- Barberis, Nicholas, Andrei Shleifer, and Robert Vishny, 1998, A model of investor sentiment, Journal of Financial Economics 49, 307–343.
- Berk, Jonathan B, Richard C. Green, and Vasant Naik, 1999, Optimal investment, growth options, and security returns, *Journal of Finance* 54, 1153–1607.
- Bernard, Victor L., and Jacob K. Thomas, 1989, Post-earnings-announcement drift: Delayed price response or risk premium? Journal of Accounting Research, Supplement 27, 1–48.
- Brav, Alon, Reuven Lehavy, and Roni Michaely, 2005, Using expectations to testing asset pricing models, *Financial Management* Autumn, 5–37.
- Campbell, John Y., Jens Hilscher, and Jan Szilagyi, 2008, In search of distress risk, Journal of Finance 63, 2899–2939.
- Chan, Louis K. C., Narasimhan Jegadeesh, and Josef Lakonishok, 1996, Momentum strategies, Journal of Finance, 51, 1681–1713.
- Chava, Sudheer and Amiyatosh Purnanandam, 2010, Is default-risk negatively related to stock returns? *Review of Financial Studies* 23, 2523–2559.
- Chen, Long, Ralitsa Petkova, and Lu Zhang, 2008, The expected value premium, *Journal of Financial Economics* 87, 269–280.
- Chen, Long, Robert Novy-Marx, and Lu Zhang, 2010, An alternative three-factor model, working paper, Ohio State University.
- Claus, James and Jacob Thomas, 2001, Equity premia as low as three percent? Evidence from analysts' earnings forecasts for domestic and international stock markets, *Journal of Finance* 56, 1629–1666.
- Cochrane, John H., 1991, Production-based asset pricing and the link between stock returns and economic fluctuations, *Journal of Finance* 46, 209–237.
- Cochrane, John H., 1996, A cross-sectional test of an investment-based asset pricing model, Journal of Political Economy 104, 572–621.
- Cooper, Michael J., Huseyin Gulen, and Michael J. Schill, 2008, Asset growth and the cross-section of stock returns, *Journal of Finance* 63, 1609–1651.
- Daniel, Kent, David Hirshleifer, and Avanidhar Subrahmanyam, 1998, Investor psychology and security market under- and overreaction, *Journal of Finance* 53, 1839–1885.
- Daniel, Kent, and Sheridan Titman, 2006, Market reactions to tangible and intangible information, Journal of Finance 61, 1605–1643.

Dichev, Ilia, 1998, Is the risk of bankruptcy a systematic risk? Journal of Finance 53, 1141–1148.

- Easton, Peter D., 2006, Use of forecasts of earnings to estimate and compare cost of capital across regimes, *Journal of Business Finance and Accounting* 33, 374–394.
- Easton, Peter D., 2007, Estimating the cost of capital implied by market prices and accounting data, *Foundations and Trends in Accounting* 2, 241–364.
- Easton, Peter D. and Steven J. Monahan, 2005, An evaluation of accounting-based measures of expected returns, *The Accounting Review* 80, 501–538.
- Easton, Peter D. and Gregory A. Sommers, 2007, Effect of analysts' optimism on estimates of the expected rate of return implied by earnings forecasts, *Journal of Accounting Research* 45, 983–1015.
- Easton, Peter D., Gary Taylor, Pervin Shroff, and Theodore Sougiannis, 2002, Using forecasts of earnings to simultaneously estimate growth and the rate of return on equity investment, *Journal of Accounting Research* 40, 657–676.
- Fama, Eugene F. and Kenneth R. French, 1993, Common risk factors in the returns on stocks and bonds, Journal of Financial Economics 33, 3–56.
- Fama, Eugene F. and Kenneth R. French, 1997, Industry costs of equity, Journal of Financial Economics 43, 153–193.
- Fama, Eugene F., and Kenneth R. French, 2006, Profitability, investment, and average returns, Journal of Financial Economics 82, 491–518.
- Fama, Eugene F. and Kenneth R. French, 2008, Dissecting anomalies, Journal of Finance 63, 1653–1678.
- Fama, Eugene F. and James D. MacBeth, 1973, Risk, return, and equilibrium: Empirical tests, Journal of Political Economy 81, 607–636.
- Francis, Jennifer, Ryan LaFond, Per M. Olsson, and Katherine Schipper, 2004, Costs of equity and earnings attributes, *The Accounting Review* 79, 967–1010.
- Gebhardt, William R., Charles M. C. Lee, and Bhaskaram Swaminathan, 2001, Toward an implied cost of capital, *Journal of Accounting Research* 39, 135–176.
- Gode, Dan, and Partha Mohanram, 2003, Inferring the cost of capital using the Ohlson-Juettner model, *Review of Accounting Studies* 8, 399–431.
- Guay Wayne, S.P. Kothari, and Susan Shu, 2005, Properties of implied cost of capital using analysts' forecasts, working paper, MIT.
- Hansen, Lars Peter, John C. Heaton, and Nan Li, 2005, Intangible risk, in Carol Corrado, John Haltiwanger, and Daniel Sichel, eds. *Measuring Capital in the New Economy*, The University of Chicago Press.
- Hong, Harrison and Jeremy C. Stein, 1999, A unified theory of underreaction, momentum trading, and overreaction in asset markets, *Journal of Finance* 54, 2143–2184.

- Hou, Kewei, Mathijs A. van Dijk, and Yinglei Zhang, 2009, The implied cost of capital: A new approach, working paper, Ohio State University.
- Ikenberry, David, Josef Lakonishok, and Theo Vermaelen, 1995, Market underreaction to open market share repurchases, Journal of Financial Economics 39, 181–208.
- Jegadeesh, Narasimhan, and Sheridan Titman, 1993, Returns to buying winners and selling losers: Implications for stock market efficiency, *Journal of Finance* 48, 65–91.
- Lakonishok, Josef, Andrei Shleifer, and Robert W. Vishny, 1994, Contrarian investment, extrapolation, and risk, *Journal of Finance* 49, 1541–1578.
- Lee, Charles M. C., David Ng, and Bhaskaran Swaminathan, 2009, Testing international asset pricing models using implied costs of capital, *Journal of Financial and Quantitative Analysis* 44, 307–335.
- Lee, Charles M. C., Eric C. So, and Charles C. Y. Wang, 2010, Evaluating implied cost of capital estimates, working paper, Stanford University.
- Li, Erica X. N., Dmitry Livdan, and Lu Zhang, 2009, Anomalies, Review of Financial Studies 22, 4301–4334.
- Liu, Laura Xiaolei, Toni M. Whited, and Lu Zhang, 2009, Investment-based expected stock returns, Journal of Political Economy 117, 1105–1139.
- Liu, Laura Xiaolei, and Lu Zhang, 2010, Investment-based momentum profits, working paper, Ohio State University.
- Loughran, Tim and Jay R. Ritter, 1995, The new issues puzzle, Journal of Finance 50, 23–51.
- Lyandres, Evgeny, Le Sun, and Lu Zhang, 2008, The new issues puzzle: Testing the investmentbased explanation, *Review of Financial Studies* 21, 2825–2855.
- Michaely, Roni, Richard H. Thaler, and Kent L. Womack, 1995, Price reactions to dividend initiations and omissions: overreaction or drift? *Journal of Finance* 50, 573–608.
- O'Brien, Patricia C., 1988, Analysts' forecasts as earnings expectations, *Journal of Accounting* and Economics 10, 53–83.
- O'Hanlon, John, and Anthony Steele, 2000, Estimating the equity risk premium using accounting fundamentals, *Journal of Business Finance and Accounting* 27, 1051–1083.
- Pastor, Lubos, Meenakshi Sinha, and Bhaskaran Swaminathan, 2008, Estimating the intertemporal risk-return tradeoff using the implied cost of capital, *Journal of Finance* 63, 2859–2897.
- Ritter, Jay R., 1991, The long-run performance of initial public offerings, *Journal of Finance* 46, 3–27.
- Rosenberg, Barr, Kenneth Reid, and Ronald Lanstein, 1985, Persuasive evidence of market inefficiency, *Journal of Portfolio Management* 11, 9–11.
- Sloan, Richard G., 1996, Do stock prices fully reflect information in accruals and cash flows about future earnings? The Accounting Review 71, 289–315.

- Titman, Sheridan, K. C. John Wei, and Feixue Xie, 2004, Capital investments and stock returns, Journal of Financial and Quantitative Analysis 39, 677–700.
- Wu, Jin (Ginger), Lu Zhang, and X. Frank Zhang, 2010, The q-theory approach to understanding the accrual anomaly, *Journal of Accounting Research* 48, 177–223.

Zhang, Lu, 2005, The value premium, Journal of Finance 60, 67–103.

A **A Brief Review of Explanations of Capital Markets Anomalies**

Explanations of capital markets anomalies can be broadly categorized into two groups. Behavioral finance argues that investors make systematic mistakes in pricing assets, and that these mistakes give rise to predictable pricing errors manifested as anomalies. New classical finance argues that risk and expected returns vary systematically with firm characteristics, and that time-varying expected returns give rise to anomalous empirical relations between average returns and firm characteristics.

A decomposition of average realized returns into expected returns and unexpected returns can disentangle the two competing schools of thought. In behavioral models, average realized returns should equal average unexpected returns, whereas in new classical models, average realized returns should equal average expected returns.

A.1 Behavioral Finance

To focus on predictable pricing errors, behavioral models typically shut down the channel of timevarying expected returns by assuming constant discount rates. Barberis, Shleifer, and Vishny (1998) propose a model of investor sentiment. In the model investors exhibit two types of psychological biases, conservatism and representative heuristics. Conservatism means that investors are slow in updating their beliefs in the face of new evidence. Barberis et al. argue that conservatism can explain the underreaction evidence that security prices underreact to news over short horizons between one to 12 months as in earnings momentum and price momentum. Representative heuristics means that after a consistent history of earnings growth over several years, investors might wrongfully believe that the past history is representative of future growth prospects. These investors then overreact to past news and send stock prices to unsustainable levels. Barberi et al. argue that this bias can explain the overreaction evidence that stocks that have had a long record of good news tend to become overpriced and have low average subsequent returns, and that stocks that have had a long record of bad news tend to become underpriced and have high average subsequent returns.

Daniel, Hirshleifer, and Subrahmanyam (1998) develop a model of investor overconfidence. Overconfidence means that investors overestimate the precision of their private information signals, but not public information signals. Overconfident investors tend to overweight their private signals relative to their prior, and cause the stock price to overreact. Self-attribution means that individuals too strongly attribute events confirming the validity of their prior actions to high ability and disconfirming events to extent noise or sabotage. When investors exhibit self-attributive overconfidence, new public signals are viewed on average as confirming the validity of their private signals, and cause continuous overreaction. The continuous overreaction explains the underreaction evidence, whereas eventual correction in the stock price explains the overreaction evidence.

Hong and Stein (1999) construct a model populated by two sets of boundedly rational investors, news-watchers and momentum traders. News-watchers observe private information, but fail to extract each other's information from prices. Coupled with slow information diffusion across the population, the stock price underreacts in the short run. Momentum traders exploit this effect by trend chasing, but their activity eventually leads to overreaction in the stock price at long horizons.

A.2 New Classical Finance

New classical models focus on time-varying expected returns as the driving force of anomalies. By retaining the assumption of rational expectations, these models shut down the channel of predictable unexpected returns. Building on Cochrane (1991, 1996) and Berk, Green, and Naik (1999), a stream of recent papers by Zhang (2005), Lyandres, Sun, and Zhang (2008), Li, Livdan, and Zhang (2009), Liu, Whited, and Zhang (2009), Chen, Novy-Marx, and Zhang (2010), Liu and Zhang (2010), Wu, Zhang, and Zhang (2010) elaborate a unified conceptual framework for understanding capital markets anomalies. In particular, Zhang argues that because of costly reversibility (higher costs in cutting than in expanding the scale of productive assets), value firms are less flexible than growth firms in scaling down to mitigate the impact of negative shocks. Because value firms have less profitable assets than growth firms, value firms want to disinvest more, especially in recessions. But because disinvesting is more costly, the cash flows of value firms are more adversely affected by bad economic conditions than the cash flows of growth firms.

Based on the first principles of firms' optimal investment decision, Liu, Whited, and Zhang (2010) argue that expected stock returns are related to the ratio of expected next-period marginal benefits of investment over current-period marginal costs of investment. Stocks with high book-to-market, low investment-to-assets, low equity issues, and low asset growth earn higher expected returns because their low investment levels imply low current-period marginal costs of investment. Intuitively, all else equal, investment and the discount rate (the expected return) are negatively correlated. Firms with low discount rates have more projects with positive net present value and consequently invest more than firms with high discount rates. Lyandres, Sun, and Zhang (2008) and Li, Livdan, and Zhang (2009) use this logic to explain the new issues puzzle of Ritter (1991) and Loughran and Ritter (1995). Treating accruals as working capital investment, Wu, Zhang, and Zhang (2010) apply this logic to explain Sloan's (1996) accrual anomaly. This discount rate channel makes sense of evidence interpreted as overreaction in behavioral finance.

In addition, stocks with high earnings surprises, high short-term prior returns, low financial distress, and high return on assets have higher expected next-period marginal benefits of investment and consequently higher expected stock returns than stocks with low earnings surprises, low shortterm prior returns, high financial distress, and low return on assets, respectively. In particular, Chen, Novy-Marx, and Zhang (2010) show that controlling for profitability (a measure of expected marginal product of capital) largely explains Campbell, Hilscher, and Szilagyi's (2008) distress anomaly. Liu, Whited, and Zhang (2009) and Liu and Zhang (2010) show that expected growth of investment-to-capital and expected marginal product of capital are quantitatively important for explaining earnings momentum and price momentum, respectively. This cash flow channel makes sense of evidence interpreted as underreaction in behavioral finance.

B Variable Definitions

B/M is the book equity at the fiscal yearend divided by the market equity in December. The book equity is the stockholders' equity (Compustat annual item SEQ), minus preferred stock, plus balance sheet deferred taxes and investment tax credit (item TXDITC) if available, minus postretirement benefit asset (item PRBA) if available. If stockholder's equity value if missing, we use common equity (item CEQ) plus preferred stock par value (item PSTK). Preferred stock is preferred stock liquidating value (item PSTKL) or preferred stock redemption value (item PSTKRV) or preferred stock par value (item PSTK) in that order of availability. If these variable are missing, we use book assets (item AT) minus liabilities (item LT). The market equity (ME) is price per share times shares outstanding from CRSP.

The five-year composite issuance (CI) measure from Daniel and Titman (2006) is defined as:

$$\iota(t-\tau) = \log\left(\frac{ME_t}{ME_{t-\tau}}\right) - r(t-\tau, t),\tag{A1}$$

where $r(t-\tau, t)$ is the cumulative log return on the stock from the last trading day of calendar year t-6 to the last trading day of calendar year t-1 and ME_t ($ME_{t-\tau}$) is total market equity on the last trading day of calendar year t (t-6) from CRSP. In economic terms, $\iota(t-\tau)$ measures the part of firm growth in market equity that is not due to stock returns. This measure is not affected by corporate decisions such as splits and stock dividends. However, issuance activities such as new equity issues, employee stock options, or any other actions that trade ownership for cash or services increase the composite issuance. In contrast, repurchase activities such as open market share repurchases, dividends, or any other action that pays cash out of a firm decrease the composite issuance.

The net stock issues (NSI) are the annual change in the logarithm of the number of real shares outstanding, which adjusts for distribution events such as splits and rights offerings. Following Fama and French (2008), we construct the net stock issues measure using the natural log of the ratio of the split-adjusted shares outstanding at the fiscal year end in t-1 divided by the splitadjusted shares outstanding at the fiscal year end in t-2. The split-adjusted shares outstanding is shares outstanding (Compustat annual item CSHO) times the adjustment factor (item ADJEX_C). If the Compustat shares or adjustment factors for calculating net stock issues are missing, we set the measure to be zero. NSI calculated in this way can be positive or negative.

Following Titman, Wei, and Xie (2004), we measure abnormal investment, AI, that applies for the portfolio formation year t, as:

$$AI_{t-1} \equiv \frac{CE_{t-1}}{(CE_{t-2} + CE_{t-3} + CE_{t-4})/3} - 1$$
(A2)

in which CE_{t-1} is capital expenditure (Compustat annual item CAPX) scaled by its sales (item SALE) in year t-1. The last three-year average capital expenditure aims to project the benchmark investment at the portfolio formation year. Using sales as the deflator assumes that the benchmark investment grows proportionately with sales. Asset growth, AG, for the portfolio formation year t is defined as the percentage change in total assets (Compustat annual item AT) from fiscal year ending in calendar year t-2 to fiscal year ending in calendar year t-1.

Following Sloan (1996), we measure total accruals, AC, for the last fiscal year ending in calendar year t-1 as changes in non-cash working capital minus depreciation expense scaled by average total assets, which is the mean of the total assets (Compustat annual item AT) for the fiscal years ending in t-1 and t-2. The non-cash working capital is the change in non-cash current assets minus the change in current liabilities less short-term debt and taxes payable.

$$TA \equiv (\triangle CA - \triangle CASH) - (\triangle CL - \triangle STD - \triangle TP) - DEP, \tag{A3}$$

in which $\triangle CA$ is the change in current assets (item ACT), $\triangle CASH$ is the change in cash or cash equivalents (item CHE), $\triangle CL$ is the change in current liabilities (item LCT), $\triangle STD$ is the change in debt included in current liabilities (item DLC), $\triangle TP$ is the change in income taxes payable (item TXP), and *DEP* is depreciation and amortization expense (item DP).

Campbell, Hilscher, and Szilagyi (2008, the third column in Table 4) measure a firm's failure probability (FP) as $1/[1 + \exp(-\text{Distress}_t)]$, in which the distress measure is constructed as:

$$Distress_{t} = -9.164 - 20.264 NIMTAAVG_{t} + 1.416 TLMTA_{t} - 7.129 EXRETAG_{t} + 1.411 SIGMA_{t} - 0.045 RSIZE_{t} - 2.132 CASHMTA_{t} + 0.075 MB_{t} - 0.058 PRICE_{t}$$
(A4)

where

$$NIMTAAVG_{t-1,t-12} \equiv \frac{1-\phi^3}{1-\phi^{12}} (NIMTA_{t-1,t-3} + \dots + \phi^9 NIMTA_{t-10,t-12})$$
$$EXRETAVG_{t-1,t-12} \equiv \frac{1-\phi}{1-\phi^{12}} (EXRET_{t-1} + \dots + \phi^{11}EXRET_{t-12})$$

The coefficient $\phi = 2^{-1/3}$ means that the weight is halved each quarter. NIMTA is net income (Compustat quarterly item NIQ) divided by the sum of market equity and total liabilities (item LTQ). The moving average NIMTAAVG is designed to capture the idea that a long history of losses is a better predictor of bankruptcy than one large quarterly loss in a single month. $EXRET = \log(1 + R_{it}) - \log(1 + R_{S\&P 500,t})$ is the monthly log excess return on each firms equity relative to the S&P 500 index. The moving average EXRETAVG is designed to capture the idea that a sustained decline in stock market value is a better predictor of bankruptcy than a sudden stock price decline in a single month. TLMTA is the ratio of total liabilities (item LTQ) divided by the sum of market equity and total liabilities. SIGMA is the volatility of each firm's daily stock return over the past three months. RSIZE is the relative size of each firm measured as the log ratio of its market equity to that of the S&P 500 index. CASHMTA, used to capture the liquidity position of the firm, is the ratio of cash and short-term investments (item CHEQ) divided by the sum of market equity and total liabilities. MB is the market-to-book equity. PRICE is the log price per share of the firm. We also winsorize the market-to-book ratio and all other variables in the construction of F-prob at the 5th and 95th percentiles of their pooled distributions across all firm-months. Finally, we winsorize PRICE at \$15.

C Measuring Portfolio Dividend Growth Rates

We measure portfolio dividend growth using returns with and without dividends, following Hansen, Heaton, and Li (2005) and Chen, Petkova, and Zhang (2008). Consider portfolios that are annually rebalanced. To describe our procedure precisely, we introduce additional notations: P_t = market equity value at the end of June for year t of the stocks allocated to the portfolio when formed at the end of June for year t; $P_{t,t+1}$ = market equity value at the end of June for year t + 1 of the stocks allocated to the portfolio at the end of June for year t; $D_{t,t+1}$ = dividends paid between portfolio formation of year t and t + 1 on the stocks allocated to the portfolio at year t; $R_{t,t+1}$ = return with dividends at the end of June of year t + 1 on a portfolio formed in year t; $G_{t,t+1}^P$ = return without dividends (rate of capital gain) observed at the end of June for year t + 1 on a portfolio formed in year t. When there are two time subscripts on a variable, the first subscript indicates the time when the portfolio is formed and the second subscript gives the time when the variable is observed. P_t can be a shorthand for $P_{t,t}$ as the market value of equity of a portfolio when formed in year t.

For each portfolio, we construct the dividend yield, $D_{t,t+1}/P_t$, from the value-weighted realized portfolio returns with and without dividends:

$$\frac{D_{t,t+1}}{P_t} = R_{t,t+1} - G_{t,t+1}^P.$$
(A5)

Because monthly total returns are compounded to get annual returns in CRSP, the dividend yield includes dividends and the reinvestment returns earned from the time a dividend is paid to the end of the annual return period. We measure portfolio dividend growth rates as:

$$G_{t+1} = \left(\frac{D_{t,t+1}/P_t}{D_{t-1,t}/P_{t-1}}\right) \left(G_{t-1,t}^P + 1\right) - 1.$$
(A6)

Because the right-hand side of equation (A6) equals $\left(\frac{D_{t,t+1}/P_t}{D_{t-1,t}/P_{t-1}}\right) \left(\frac{P_{t-1,t}}{P_{t-1}}\right) - 1$, the equation says that the dividend growth rate is (dividends at t + 1 per dollar invested at t multiplied by dollars invested at t)/(dividends at t per dollar invested at t - 1 multiplied by dollars invested at t - 1). The reinvested capital gain embedded in equation (A6), $P_{t-1,t}/P_{t-1}$, is important: high $P_{t-1,t}/P_{t-1}$ means more dollars to invest at t and higher dividend growth rates.

For monthly rebalanced momentum, SUE, and ROA portfolios, we aggregate monthly portfolio returns with and without dividends from July of year t to June of year t + 1 to annual returns with and without dividends for year t. We then apply equations (A5) and (A6) on the aggregated annual returns with and without dividends to construct annual dividend growth rates for the portfolios. Aggregating over monthly returns with and without dividends to obtain annual returns with and without dividends alleviates the effect of dividend seasonality on the calculation of portfolio dividend growth rates. For the failure probability portfolios, monthly observations of returns are already one-year buy-and-hold returns. As such, we apply equations (A5) and (A6) directly on the monthly observations of returns to construct dividend growth rates for these portfolios.

Table 1 : Descriptive Statistics, Samples for Estimating Implied Costs of Equity

We present descriptive statistics including the mean, standard deviation, min, 25% percentile, median, 75% percentile, and max for all the anomaly variables. We also report the sample period and average number of firms in the cross-section for each sample that corresponds to a given anomaly variable. Book-to-market (B/M) is the book equity divided by the market equity at the end of fiscal year, and the book equity is measured as in Fama and French (1993). Size (ME)is market capitalization in millions of dollars. Composite issuance (CI) is the cumulative log five-year growth rate of total market equity minus the cumulative log five-year stock return. Net stock issues (NSI) are the natural log of the ratio of the split-adjusted shares outstanding at the fiscal year ending in calendar year t-1 divided by the split-adjusted shares outstanding at the fiscal year ending in calendar year t-2. Abnormal investment (AI) is the deviation of the current year investment-to-sales ratio from the past three-year moving average investment-to-sales. Asset growth (AG) is the percentage change in total assets from the fiscal year ending in calendar year t-2 to the fiscal year ending in calendar year t-1. Investment-to-assets (I/A) is the annual change in property, plant, and equipment plus the annual change in inventory divided by lagged total assets. Accruals (AC) are changes in non-cash working capital minus depreciation expense (scaled by average total assets) as in Sloan (1996). Earnings surprise (SUE) is the unexpected earnings defined as the most recent quarterly earnings per share minus earnings per share four quarters ago divided by the standard deviation of the unexpected earnings from the prior eight quarters. The distress measure is constructed as in Compbell, Hilscher, and Szilagyi (2008) and the failure probability (*FP*, in percent) is calculated as $1/[1 + \exp(-\text{Distress})]$. Return-on-assets (*ROA*) is the most recent earnings divided by one-quarter-lagged total assets. Past five-year sales growth (SG)is the sales growth from year t-5 to t. Prior returns (MOM) are prior six-month returns at each portfolio formation month. See Section 2.1 and Appendix B for detailed variable definitions.

	Sample	# Firms	Mean	Std	Min	25%	50%	75%	Max						
		Pan	el A: The	baseline in	nplied cos	ts of equi	ity estima	ation							
B/M	80-08	2201	1.51	5.48	0.07	0.40	0.66	1.01	59.67						
ME	80-08	2201	2147.63	6225.35	9.53	132.02	413.53	1368.66	57017.85						
CI	80-08	1393	0.00	0.41	-1.67	-0.19	-0.05	0.16	1.70						
NSI	80-08	2200	0.04	0.10	-0.22	0.00	0.01	0.03	0.65						
AI	80-08	1513	0.29	0.51	-0.81	0.06	0.21	0.41	3.77						
AG	80-08	1812	0.18	0.36	-0.40	0.01	0.09	0.22	2.66						
I/A	80 - 08	1912	0.10	0.17	-0.36	0.02	0.07	0.14	1.13						
AC	80 - 08	1631	-0.03	0.08	-0.32	-0.07	-0.04	0.01	0.30						
SUE	80 - 08	2006	-0.10	3.41	-77.89	-0.63	0.05	0.67	37.06						
FP	80-08	2038	0.06	0.13	0.01	0.03	0.04	0.06	2.88						
ROA	80-08	2161	0.04	0.12	-0.64	0.00	0.04	0.08	0.40						
MOM	80-08	2317	0.08	0.33	-0.80	-0.10	0.05	0.22	3.79						
	Panel B: The modified implied costs of equity estimation $(\text{the full Ferrer French } POE \text{ forecasting more spin})$														
	(the full Fama-French <i>ROE</i> forecasting regression)														
B/M	75 - 08	2091	1.41	3.38	0.11	0.50	0.82	1.28	34.44						
ME	75 - 08	2091	1073.27	2792.25	3.10	45.07	174.47	723.69	22248.26						
CI	75 - 08	1134	-0.05	0.43	-1.84	-0.22	-0.07	0.12	1.61						
NSI	75 - 08	2091	0.03	0.10	-0.24	0.00	0.00	0.02	0.69						
AI	75 - 08	1540	0.26	0.48	-0.77	0.04	0.18	0.37	3.58						
AG	75 - 08	2091	0.14	0.29	-0.35	0.00	0.08	$\begin{array}{c} 0.19\\ 0.14\end{array}$	1.94						
I/A	75 - 08	2076	0.09	0.16	-0.37	0.02			0.96						
AC	75 - 08	1951	-0.03	0.08	-0.31	-0.07	-0.03	0.01	0.29						
SUE	77-08	2119	0.48	29.41	-63.94	-0.59	0.07	0.68	1369.95						
FP	75–08	2121	0.07	0.14	0.01	0.03	0.04	0.07	2.77						
ROA	77-08	2268	0.04	0.12	-0.68	0.00	0.04	0.08	0.40						
MOM	75 - 08	3108	0.09	0.34	-0.77	-0.10	0.05	0.22	4.80						
				modified ir	-	-									
		(the	simplified	Fama-Fre	nch ROE	forecasti	ng regres	sion)							
B/M	75 - 08	2893	1.38	3.03	0.11	0.53	0.86	1.30	30.93						
ME	75 - 08	2893	1070.64	2811.91	3.08	44.16	167.96	705.14	22495.58						
CI	75 - 08	1534	-0.05	0.44	-1.86	-0.23	-0.07	0.13	1.65						
NSI	75 - 08	2891	0.03	0.10	-0.24	0.00	0.00	0.02	0.70						
AI	75 - 08	2025	0.26	0.46	-0.73	0.04	0.19	0.37	3.40						
AG	75 - 08	2396	0.14	0.29	-0.36	0.00	0.08	0.19	1.95						
I/A	75 - 08	2556	0.09	0.16	-0.38	0.01	0.06	0.13	0.98						
AC	75 - 08	2181	-0.03	0.09	-0.31	-0.07	-0.03	0.01	0.30						
SUE	77 - 08	2844	0.29	26.57	-184	-0.58	0.08	0.69	1355.05						
FP	75 - 08	2920	0.08	0.15	0.01	0.03	0.04	0.07	3.05						
ROA	77 - 08	3128	0.04	0.12	-0.68	0.00	0.04	0.08	0.40						
MOM	75 - 08	3109	0.09	0.34	-0.77	-0.10	0.05	0.22	4.67						

Table 2 : Multiple Regress	ions to Forecast Profitability
----------------------------	--------------------------------

The table shows average slopes and their Fama-MacBeth t-statistics from annual cross-sectional regressions to predict profitability, $Y_{t+\tau}/B_{t+\tau-1}$, one, two, and three years ahead ($\tau = 1, 2, 3$). Y_t, D_t , and AC_t are earnings, dividends, and accruals per share for the fiscal year ending in calendar year t. $-AC_t$ is accruals for firms with negative accruals (zero otherwise) and $+AC_t$ is accruals for firms with negative accruals (zero otherwise) and $+AC_t$ is accruals for firms with positive accruals (zero otherwise). B_t is book equity per share at the end of fiscal year t. ME_t is market capitalization (price times shares outstanding) at the end of fiscal year t. Neg Y_t is a dummy that is one for firms with negative earnings for fiscal year t (zero otherwise), and No D_t is a dummy that is one for firms that pay no dividends during fiscal year t. The sample is from 1963 to 2008. Int. is the regression intercept, and the R^2 is adjusted for degrees of freedom.

au	Int.	$\ln B_t/M_t$	$\ln ME_t$	Neg Y_t	Y_t/B_t	$-AC_t/B_t$	$+AC_t/B_t$	AG_t	No D_t	D_t/B_t	R^2		
	Panel A: The full Fama-French (2006) specification												
Ave	Average slopes												
1	0.01	-0.03	0.01	-0.04	0.63	-0.10	-0.03	-0.03	-0.02	0.12	0.43		
2	0.00	-0.02	0.01	-0.07	0.39	-0.09	0.01	-0.05	-0.02	0.38	0.21		
3	0.01	-0.01	0.01	-0.07	0.27	-0.09	0.02	-0.05	-0.02	0.52	0.13		
<i>t</i> -st	atistic	s											
1	0.67	-4.09	3.26	-2.69	18.39	-5.80	-2.98	-4.60	-4.71	2.61			
2	0.15	-2.65	3.18	-3.59	13.45	-3.51	0.42	-6.61	-4.70	8.10			
3	0.29	-1.96	3.26	-3.27	9.67	-5.16	1.14	-5.60	-4.32	12.34			
			Panel	B: The s	implifie	ed Fama-Fre	ench specifi	cation					
Ave	erage s	slopes											
1	0.00	-0.02	0.01	-0.05	0.61			-0.04			0.43		
2	-0.00	-0.02	0.01	-0.07	0.40			-0.06			0.20		
3	0.00	-0.01	0.01	-0.06	0.31			-0.06			0.13		
<i>t</i> -st	atistic	s											
1	0.23	-4.25	4.26	-2.88	18.33			-5.76					
2	-0.09	-2.62	4.86	-3.48	11.25			-8.06					
3	0.05	-1.78	5.15	-3.10	9.85			-8.72					

Table 3: Average Realized Returns and Expected Returns, Implied Costs of Equity, Baseline and Modified

We report the average realized returns, A[R], the implied costs of equity from the baseline residual income model that uses the forecasted earnings from IBES, $E_0[R]$, the implied costs of equity from the modified residual income model that uses the Fama-French (2006) forecasted ROE, $E_1[R]$, and the implied costs of equity from the modified residual income model that uses the simplified Fama-French forecasted ROE, $E_2[R]$. In June of each year t from 1980 to 2008, we sort all NYSE stocks on book-to-market (B/M), size (ME). composite issuance (CI), net stock issues (NSI), abnormal investment (AI), asset growth (AG), investment-to-assets (I/A), and total accruals (AC) for the fiscal year ending in calendar year t-1 and use the NYSE breakpoints to split NYSE, Amex, and Nasdaq stocks into five quintiles. Value-weighted portfolio returns are calculated from July of year t to June of year t+1. We also sort all NYSE stocks each month on the prior six-month returns (MOM) and earnings surprises (SUE), and use the NYSE breakpoints to split all stocks into quintiles. We hold the portfolios for six months and calculate value-weighted returns. Each month we use NYSE/Amex/Nasdag breakpoints to sort all stocks on Campbell, Hilscher, and Szilagzi's (2008) failure probability (FP) into quintiles and calculate one-year value-weighted returns for each portfolio. Each month we also use NYSE breakpoints to sort all stocks on quarterly return-on-assets (ROA) and calculate value-weighted returns for the current month. Earnings and other Compustat quarterly accounting data for a fiscal quarter are used in portfolio sorts in the months immediately after its public earnings announcement month (Compustat quarterly item RDQ). See Section 2.1 and Appendix B for detailed variable definitions. "H-L" is the high-minus-low portfolios and "[t]" is heteroscedasticity-and-autocorrelation-consistent t-statistics testing a given H-L moment is zero. The sample periods are described in Table 1. All entries other than [t] are in annualized percent.

	A[R]	$E_0[R]$	$E_1[R]$	$E_2[R]$	A[R]	$E_0[R]$	$E_1[R]$	$E_2[R]$	A[R]	$E_0[R]$	$E_1[R]$	$E_2[R]$	A[R]	$E_0[R]$	$E_1[R]$	$E_2[R]$		
		Panel A	anel A: B/M Panel B: ME							Panel	$\mathbf{C}:CI$		Panel D: NSI					
Low	12.1	8.6	7.2	7.5	15.8	12.7	10.9	11.3	15.1	10.4	10.1	10.6	16.1	10.3	9.4	10.0		
3	14.8	11.0	10.1	10.8	14.2	11.1	10.0	10.4	13.4	9.8	9.1	9.5	13.1	9.8	9.0	9.5		
High	17.3	14.9	15.7	16.0	12.8	9.6	8.7	9.4	10.9	10.3	9.0	9.8	8.6	10.2	8.9	9.6		
H-L	5.2	6.3	8.5	8.5	-3.0	-3.1	-2.2	-1.8	-4.2	-0.1	-1.1	-0.8	-7.5	-0.1	-0.5	-0.5		
[t]	2.2	12.0	8.4	9.2	-0.8	-7.6	-6.0	-5.3	-2.8	-0.2	-4.2	-2.6	-3.0	-0.4	-1.6	-1.9		
	Panel E: AI Panel F: AG								Panel	G: I/A		Panel H: AC						
Low	15.0	10.0	9.1	9.4	16.1	10.2	9.6	9.8	14.2	10.5	9.6	9.9	13.9	9.9	8.9	9.1		
3	14.2	9.9	9.6	10.2	13.2	9.7	9.6	10.0	13.2	9.7	9.3	9.7	14.4	9.8	9.1	9.4		
High	11.1	9.9	8.1	8.9	10.5	9.6	8.1	8.5	10.8	9.9	8.6	9.1	9.1	9.8	8.4	8.8		
H-L	-4.0	-0.1	-1.0	-0.5	-5.6	-0.6	-1.5	-1.2	-3.4	-0.5	-1.0	-0.9	-4.8	-0.0	-0.5	-0.3		
[t]	-2.4	-0.2	-2.7	-1.1	-2.8	-3.0	-5.6	-4.4	-1.8	-3.0	-5.6	-4.5	-3.6	-0.1	-3.3	-1.8		
		Panel	I: SUE			Panel	J: FP			Panel K: ROA				Panel L: MOM				
Low	8.6	10.1	9.2	9.9	13.8	9.2	7.8	8.2	5.9	11.0	9.8	10.6	6.8	11.3	10.0	10.7		
3	10.9	10.1	8.9	9.5	12.4	11.2	9.8	10.4	10.1	10.5	10.0	10.3	11.2	10.2	9.0	9.7		
High	13.5	10.0	8.3	8.9	5.7	13.1	10.8	11.5	12.4	9.3	7.7	8.0	13.2	9.7	7.8	8.4		
H-L	4.9	-0.1	-0.9	-1.0	-8.1	3.8	3.0	3.3	6.5	-1.7	-2.1	-2.7	6.4	-1.7	-2.2	-2.3		
[t]	5.5	-3.1	-19.2	-18.5	-5.0	34.8	21.0	25.8	3.3	-22.0	-19.3	-35.1	3.3	-17.1	-23.7	-23.5		

Table 4 : Descriptive Statistics, Samples for Estimating Expected Returns and Expected Growth Rates Simultaneously

We present descriptive statistics including the mean, standard deviation, min, 25% percentile, median, 75% percentile, and max for all the anomaly variables. We also report the sample period and average number of firms in the cross-section for each sample that corresponds to a given anomaly variable. Book-to-market (B/M) is the book equity divided by the market equity at the end of fiscal year, and the book equity is measured as in Fama and French (1993). Size (ME)is market capitalization in millions of dollars. Composite issuance (CI) is the cumulative log five-year growth rate of total market equity minus the cumulative log five-year stock return. Net stock issues (NSI) are the natural log of the ratio of the split-adjusted shares outstanding at the fiscal year ending in calendar year t-1 divided by the split-adjusted shares outstanding at the fiscal year ending in calendar year t-2. Abnormal investment (AI) is the deviation of the current year investment-to-sales ratio from the past three-year moving average investment-to-sales. Asset growth (AG) is the percentage change in total assets from the fiscal year ending in calendar year t-2 to the fiscal year ending in calendar year t-1. Investment-to-assets (I/A) is the annual change in property, plant, and equipment plus the annual change in inventory divided by lagged total assets. Accruals (AC) are changes in non-cash working capital minus depreciation expense (scaled by average total assets) as in Sloan (1996). Earnings surprise (SUE) is the unexpected earnings defined as the most recent quarterly earnings per share minus earnings per share four quarters ago divided by the standard deviation of the unexpected earnings from the prior eight quarters. The distress measure is constructed as in Compbell, Hilscher, and Szilagyi (2008) and the failure probability (*FP*, in percent) is calculated as $1/[1 + \exp(-\text{Distress})]$. Return-on-assets (*ROA*) is the most recent earnings divided by one-quarter-lagged total assets. Past five-year sales growth (SG)is the sales growth from year t-5 to t. Prior returns (MOM) are prior six-month returns at each portfolio formation month. See Section 2.1 and Appendix B for detailed variable definitions.

	Sample	# Firms	Mean	Std	Min	25%	50%	75%	Max						
			Pane	l A: The b	aseline ET	SS estima	ation								
B/M	80-08	3026	1.58	6.59	0.05	0.37	0.63	0.98	79.12						
$\stackrel{\prime}{ME}$	80-08	3026	1899.10	5580.70	8.92	127.32	368.64	1195.65	50932.74						
CI	80-08	1649	0.01	0.43	-1.68	-0.19	-0.04	0.18	1.76						
NSI	80-08	2777	0.05	0.12	-0.22	0.00	0.01	0.04	0.80						
AI	80-08	1753	0.32	0.58	-1.05	0.06	0.22	0.43	4.37						
AG	80-08	2296	0.22	0.46	-0.42	0.01	0.10	0.25	3.58						
I/A	80-08	2415	0.11	0.19	-0.36	0.02	0.07	0.15	1.32						
AC	80-08	2061	-0.03	0.09	-0.33	-0.07	-0.03	0.01	0.33						
SUE	80-08	2442	-0.10	3.42	-84.30	-0.63	0.05	0.66	39.43						
FP	80-08	2222	0.07	0.15	0.01	0.03	0.04	0.06	3.41						
ROA	80 - 08	2560	0.04	0.12	-0.80	0.00	0.04	0.08	0.40						
MOM	80-08	2710	0.08	0.34	-0.82	-0.11	0.05	0.22	4.13						
	Panel B: The modified ETSS estimation														
	(the full Fama-French ROE forecasting regression)														
B/M	75 - 08	2851	1.43	3.73	0.10	0.48	0.79	1.24	40.52						
ME	75 - 08	2851	959.22	2509.42	2.64	46.09	164.44	645.51	20150.88						
CI	75 - 08	1507	-0.05	0.45	-1.87	-0.22	-0.06	0.14	1.72						
NSI	75 - 08	2850	0.04	0.12	-0.24	0.00	0.01	0.04	0.79						
AI	75 - 08	1859	0.27	0.53	-0.88	0.03	0.18	0.38	3.90						
AG	75 - 08	2851	0.17	0.35	-0.38	0.00	0.09	0.22	2.36						
I/A	75 - 08	2824	0.10	0.17	-0.39	0.02	0.07	0.15	1.15						
AC	75 - 08	2654	-0.03	0.09	-0.34	-0.07	-0.03	0.02	0.33						
SUE	77-08	2579	0.31	25.18	-76.97	-0.61	0.06	0.68	1334.37						
FP	75 - 08	2337	0.09	0.18	0.01	0.03	0.04	0.07	3.56						
ROA	77-08	2564	0.00	0.16	-0.88	0.00	0.04	0.08	0.40						
MOM	75 - 08	2805	0.09	0.37	-0.79	-0.12	0.04	0.23	4.64						
			Panel	C: The O	Hanlon-Ste	eele estim	ation								
B/M	65 - 08	3369	1.55	5.14	0.04	0.48	0.79	1.22	64.94						
ME	65 - 08	3369	973.60	3049.27	1.54	39.13	139.05	559.04	27092.75						
CI	65 - 08	1749	-0.05	0.41	-1.64	-0.22	-0.07	0.12	1.60						
NSI	65 - 08	3369	0.04	0.11	-0.23	0.00	0.01	0.03	0.88						
AI	65 - 08	1983	0.25	0.47	-1.04	0.04	0.17	0.35	3.48						
AG	65 - 08	2825	0.17	0.39	-0.49	0.00	0.09	0.21	3.19						
I/A	65 - 08	2973	0.10	0.20	-0.47	0.02	0.07	0.14	1.50						
AC	70 - 08	2431	-0.02	0.09	-0.41	-0.07	-0.03	0.02	0.37						
SUE	77-08	3311	0.20	23.56	-195.56	-0.60	0.07	0.69	1341.16						
FP	75 - 08	2975	0.09	0.19	0.01	0.03	0.04	0.08	3.89						
ROA	77–08	3310	0.00	0.16	-1.00	0.00	0.04	0.08	0.40						
MOM	65 - 08	3765	0.08	0.38	-0.84	-0.12	0.04	0.22	5.94						

Table 5: Expected Growth Rates, the Baseline and Modified Easton Models, the O'Hanlon-Steele Model

We report the estimated growth rates from the baseline Easton et al. (2002) model that uses the forecasted earnings from IBES, q_0 , the growth rates from the modified Easton et al. model that uses the Fama-French (2006) forecasted ROE, q_1 , and the estimated growth rates from the O'Hanlon-Steele model, g_2 . For comparison, we also report the average dividend growth rates from the dividend discounting model, A[G]. In June of each year t from 1980 to 2008, we sort all NYSE stocks on book-to-market (B/M), size (ME), composite issuance (CI), net stock issues (NSI), abnormal investment (AI), asset growth (AG), investment-to-assets (I/A), and total accruals (AC) for the fiscal year ending in calendar year t-1 and use the NYSE breakpoints to split NYSE. Amex, and Nasdaq stocks into quintiles. Value-weighted portfolio returns are calculated from July of year t to June of year t + 1. We also sort all NYSE stocks each month on the prior six-month returns (MOM) and earnings surprises (SUE), and use the NYSE breakpoints to split all stocks into quintiles. We hold the portfolios for six months and calculate value-weighted returns. Each month we use NYSE/Amex/Nasdaq breakpoints to sort all stocks on Campbell, Hilscher, and Szilagzi's (2008) failure probability (FP) into quintiles and calculate one-year value-weighted returns for each portfolio. Each month we also use NYSE breakpoints to sort all stocks on quarterly return-on-assets (ROA) and calculate value-weighted returns for the current month. Earnings and other Compustat quarterly accounting data for a fiscal quarter are used in portfolio sorts in the months immediately after its public earnings announcement month (Compustat quarterly item RDQ). See Section 2.1 and Appendix B for detailed variable definitions. "H-L" is the high-minus-low portfolios and "[t]" is heteroscedasticity-and-autocorrelation-consistent t-statistics testing a given H-L moment is zero. The sample periods are in Table 1. All entries other than [t] are in annualized percent.

	A[G]	g_0	g_1	g_2	A[G]	g_0	g_1	g_2		A[G]	g_0	g_1	g_2	A[G]	g_0	g_1	g_2		
		Panel	A: B/M	-		Panel B: ME						C: CI			Panel D: NSI				
Low	5.0	6.3	17.5	18.6	10.5	2.2	4.8	3.9		6.9	1.6	8.2	11.3	11.0	5.2	8.9	10.3		
3	7.7	5.3	10.8	10.1	7.5	13.1	8.9	11.2		5.9	6.6	8.6	9.0	9.4	5.0	9.0	11.7		
High	9.1	1.4	2.2	4.7	5.3	7.4	10.0	14.8		5.8	10.9	8.6	11.9	4.4	8.8	9.4	13.4		
H-L	4.1	-4.9	-15.4	-13.9	-5.2	5.1	5.2	10.9		-1.1	9.2	0.4	0.6	-6.6	3.6	0.5	3.1		
[t]	1.3	-2.6	-27.1	-10.7	-1.1	4.0	5.6	4.5		-0.4	3.7	0.4	0.5	-1.2	2.1	0.8	3.6		
		Pane	l E: AI			Panel F: AG					Panel (G: I/A		Panel H: AC					
Low	7.3	8.3	3.4	4.2	8.8	10.3	5.4	4.9		7.4	6.2	5.9	8.9	8.2	8.8	8.1	9.5		
3	6.5	6.1	8.5	9.2	5.8	6.8	9.7	10.6		5.7	6.9	10.0	11.6	7.3	9.3	9.9	12.0		
High	8.5	8.4	12.1	19.6	9.4	15.1	11.6	18.8		8.5	14.1	11.1	16.1	7.1	8.2	11.5	17.6		
H-L	1.2	0.1	8.6	15.5	0.6	4.8	6.2	13.9		1.1	8.0	5.2	7.3	-1.1	-0.6	3.5	8.1		
[t]	0.4	0.0	6.1	9.7	0.1	1.3	5.8	11.0		0.4	2.2	7.6	8.4	-0.3	-0.2	4.9	6.5		
		Panel	I: SUE			Panel J: FP					Panel K: ROA				Panel L: MOM				
Low	8.1	11.2	14.8	14.6	11.7	11.5	15.3	17.6		6.7	6.2	5.9	7.4	6.2	10.7	12.2	12.0		
3	8.8	12.9	12.5	11.6	9.0	12.7	11.2	12.6		7.6	10.5	11.9	11.0	9.2	11.9	13.3	12.3		
High	12.2	13.3	13.0	13.3	29.4	-8.6	-3.9	-4.4		12.1	15.7	19.1	19.8	12.6	14.0	12.7	13.3		
H-L	4.1	2.2	-1.8	-1.3	17.8	-20.0	-19.2	-22.0		5.4	9.5	13.2	12.5	6.5	3.3	0.4	1.4		
[t]	7.6	4.7	-10.1	-4.1	4.1	-15.8	-23.2	-15.3		4.6	16.1	34.0	20.9	7.3	8.3	1.7	4.6		

Table 6: Average Returns and Expected Returns, the Baseline and Modified Easton Models, the O'Hanlon-Steele Model

We report the average realized returns, A[R], the expected returns from the baseline Easton et al. (2002) model that uses the forecasted earnings from IBES, r_0 , the expected returns from the modified Easton et al. model that uses the Fama-French (2006) forecasted ROE, r_1 , and the expected returns from the O'Hanlon-Steele model, r_2 . In June of each year t from 1980 to 2008, we sort all NYSE stocks on book-to-market (B/M), size (ME), composite issuance (CI), net stock issues (NSI), abnormal investment (AI), asset growth (AG), investment-to-assets (I/A), and total accruals (AC) for the fiscal year ending in calendar year t - 1 and use the NYSE breakpoints to split NYSE, Amex, and Nasdaq stocks into quintiles. Value-weighted portfolio returns are calculated from July of year t to June of year t + 1. We also sort all NYSE stocks each month on the prior six-month returns (MOM) and earnings surprises (SUE), and use the NYSE breakpoints to split all stocks into quintiles. We hold the portfolios for six months and calculate value-weighted returns. Each month we use NYSE/Amex/Nasdaq breakpoints to sort all stocks on Campbell, Hilscher, and Szilagzi's (2008) failure probability (FP) into quintiles and calculate one-year value-weighted returns for each portfolio. Each month we also use NYSE breakpoints to sort all stocks on quarterly return-on-assets (ROA) and calculate value-weighted returns for the current month. Earnings and other Compustat quarterly accounting data for a fiscal quarter are used in portfolio sorts in the months immediately after its public earnings announcement month (Compustat quarterly item RDQ). See Section 2.1 and Appendix B for detailed variable definitions. "H-L" is the high-minus-low portfolios and "[t]" is heteroscedasticity-and-autocorrelation-consistent t-statistics testing a given H-L moment is zero. The sample periods are in Table 1. All entries other than [t] are in annualized percent.

	A[R]	r_0	r_1	r_2	A[R]	r_0	r_1	r_2	_	A[R]	r_0	r_1	r_2	A[R]	r_0	r_1	r_2		
		Panel A	A: B/M	: B/M Panel B: ME							Panel	C: CI			Panel D: NSI				
Low	12.7	10.7	18.6	19.6	15.4	4.9	5.3	5.0	-	15.3	8.1	12.4	14.6	17.0	10.5	12.4	13.2		
3	15.2	9.9	11.7	11.8	14.1	14.0	10.8	12.6		14.1	10.5	12.1	12.5	14.2	9.3	11.7	13.8		
High	16.9	8.6	5.9	7.9	13.2	11.3	12.8	16.4		12.1	13.2	10.8	14.0	9.7	11.2	10.8	14.6		
H-L	4.2	-2.1	-12.7	-9.4	-2.2	6.4	7.5	11.8		-3.2	5.2	-1.6	-1.9	-7.3	0.8	-1.7	-0.2		
[t]	1.7	-1.0	-12.2	-6.0	-0.6	5.6	5.6	4.7		-2.0	3.0	-2.1	-2.0	-2.7	0.7	-2.6	-0.2		
	Panel E: AI Panel F: AG						_	Panel G: I/A				Panel H: AC							
Low	15.0	11.8	7.2	6.7	15.9	12.1	7.8	6.6		14.4	10.9	9.1	10.7	14.3	11.7	10.3	10.7		
3	14.2	10.4	11.9	12.5	13.7	10.8	12.8	13.2		13.7	10.9	12.8	13.9	14.8	12.8	12.6	14.5		
High	12.5	11.2	14.2	20.8	11.7	16.7	13.3	19.9		11.9	15.5	12.8	17.4	10.7	11.1	13.4	19.0		
H-L	-2.5	-0.6	6.9	12.8	-4.3	4.6	5.4	12.6		-2.5	4.6	3.7	6.2	-3.6	-0.6	3.1	8.4		
[t]	-1.4	-0.2	6.4	9.7	-2.2	1.9	5.8	9.3		-1.2	1.8	6.2	7.0	-3.7	-0.3	5.1	6.9		
		Panel	I: SUE			Panel	J: FP		_	Panel K: ROA				Panel L: MOM					
Low	9.2	13.9	17.0	17.1	14.3	14.8	17.0	19.1		6.6	6.8	6.7	7.0	7.2	13.7	14.2	13.8		
3	10.9	15.9	14.5	13.6	12.6	15.3	13.0	14.3		10.4	13.6	13.6	12.5	11.5	15.3	15.6	14.5		
High	13.9	16.3	14.6	14.9	7.2	-9.4	-4.5	-4.5		12.8	18.5	20.2	21.5	13.8	15.9	14.0	14.7		
H-L	4.6	2.4	-2.5	-2.2	-7.2	-24.2	-21.6	-23.6		6.2	11.7	13.5	14.5	6.6	2.2	-0.2	0.9		
[t]	5.6	7.5	-16.6	-8.1	-4.4	-16.8	-19.6	-17.0		3.4	21.9	30.8	24.9	3.4	6.6	-0.7	3.3		